Author’s response to reviews

Title: Lifestyle modification and metformin as long-term treatment options for obese adolescents: study protocol

Authors:

A. Justine Wilson (awils29@uwo.ca)
Harry Prapavessis (hprapave@uwo.ca)
Mary E Jung (mjung32@uwo.ca)
Anita G Cramp (acramp2@uwo.ca)
Joy Vascotto (joy.vascotto@lhsc.on.ca)
Larissa Lenhardt (larissa.lenhardt@lhsc.on.ca)
J. Kevin Shoemaker (kshoemak@uwo.ca)
Margaret Watson (margaret.watson@lhsc.on.ca)
Tracy Robinson (tracy.robinson@lhsc.on.ca)
Cheril L Clarson (cheril.clarson@lhsc.on.ca)

Version: 2 Date: 20 November 2009

Author’s response to reviews:

Major revisions:
Tables - bullet points have been removed.

Minor revisions:
Methods - heading changed to Methods/design in the Abstract and manuscript.
Underlining - has been removed.
References: full citation has been provided for #29
Tables - visible vertical lines from your tables have been removed
Figures: Figure has been removed from manuscript and uploaded to this webpage. Figure legend is still in the manuscript following the references
Typography: We have completed a final read through of this paper.